Author's response to reviews

Title: Feasibility, recruitment and factors associated with motivation and satisfaction in an interventional study with 210 healthy volunteers without financial compensation.

Authors:

Quentin Luzurier (quentin.luzurier@chu-rouen.fr)
Cedric Damm (cedric.damm@chu-rouen.fr)
Fabien Lion (fabien.lion@gustaveroussy.fr)
Carine Daniel (carine.daniel@chu-rouen.fr)
Lucille Pellerin (lucille.pellerin-kurtz@chu-rouen.fr)
Marie Pierre Tavolacci (marie-pierre.tavolacci@chu-rouen.fr)

Version: 2
Date: 25 April 2014

Author's response to reviews:

Dear Editor

As you request we include line and page numbers and Authors' Contributions section after the Competing Interests section

Best regards

Marie Pierre Tavolacci